



JGR Oceans

RESEARCH ARTICLE

10.1029/2018JC014765

Key Points:

- DIN increased; PO₄-P decreased; SiO₃-Si decreased before 1987 and increased after 1987 in the Bohai Sea; winter concentrations are higher
- N/P increased, Si/N decreased, and Si/P slightly increased in the Bohai Sea; ratios are higher in winter than in summer
- Variations in Si/N ratios may have ecological impacts on the red tide features in the Bohai Sea

Supporting Information:

- Supporting Information S1
- Table S1

Correspondence to:

Q. Yao, qzhyao@ouc.edu.cn

Citation:

Wang, J., Yu, Z., Wei, Q., & Yao, Q. (2019). Long-term nutrient variations in the Bohai Sea over the past 40 years. *Journal of Geophysical Research: Oceans*, 124, 703–722. https://doi.org/ 10.1029/2018JC014765

Received 16 NOV 2018 Accepted 13 DEC 2018 Accepted article online 15 DEC 2018 Published online 28 JAN 2019

Long-Term Nutrient Variations in the Bohai Sea Over the Past 40 Years

Junjie Wang¹, Zhigang Yu^{1,2}, Qinsheng Wei^{2,3}, and Qingzhen Yao^{1,2}

¹Key Laboratory of Marine Chemistry Theory and Technology, Ministry of Education, Ocean University of China, Qingdao, China, ²Laboratory of Marine Ecology and Environmental Science, Qingdao National Laboratory for Marine Science and Technology, Qingdao, China, ³First Institute of Oceanography, State Oceanic Administration, Qingdao, China

Abstract As China's only continental sea, the Bohai Sea is a relatively closed environment and is vulnerable to natural changes and human activities. In this paper, the long-term variations in nutrients in the Bohai Sea and the potential influencing factors were analyzed based on historical summer and winter data from 1978 to 2016. The results showed that the concentrations of dissolved inorganic nitrogen (DIN) in the Bohai Sea continuously increased from 1990 and rapidly increased after 2002, the phosphate (PO₄-P) concentration exhibited a decreasing trend, and the silicate (SiO₃-Si) concentration decreased from 1978 to 1987 and increased from 1987 to 2008. The nutrient concentrations were lower in summer than in winter, and the bottom concentrations were higher than the surface concentrations in summer, whereas the vertical differences were insignificant in winter. The dominant factor determining the long-term variations in DIN were atmospheric deposition and nonpoint sources due to various human activities; the primary factors affecting PO₄-P were riverine inputs and nonpoint sources related to natural changes and human activities; the primary factors afftecting SiO₃-Si were riverine inputs. The N/P ratio followed the DIN variation, the Si/P ratio followed the SiO₃-Si variation, and the Si/N ratio decreased. The nutrient ratios were lower at the bottom than at the surface and were lower in winter than in summer. The nutrient limitation changed from nitrogen limitation to phosphorus and silicon limitations. The long-term nutrient variations in the Bohai Sea have potential ecological impacts on the local red tide features.

Plain Language Summary Nutrients are important to primary productivity and thus important to marine ecosystems. Nutrient variations in coastal oceans may potentially impact the occurrence of ecological disasters. As the only continental sea in China, the Bohai Sea is a relatively closed environment and is vulnerable to natural and anthropogenic impacts. We found that the concentration of dissolved inorganic nitrogen in the Bohai Sea increased from 1990 to 2006; the phosphate concentration decreased from 1978 to 2016, and the silicate concentration decreased before 1987 and increased after 1987. We also found that N/P increased and Si/N decreased. The nutrient concentrations were higher, whereas the nutrient ratios were lower in winter than in summer. In summer, the bottom nutrient concentrations were higher than the surface concentrations, whereas the bottom ratios were lower than the surface ratios; in winter, the vertical differences were insignificant. The primary factors driving the silicate variations were riverine inputs related to natural changes and human activities; the primary factors in the phosphate variations could be riverine inputs and nonpoint sources related to natural changes and human activities; the dominant factor in the dissolved inorganic nitrogen variations could be atmospheric deposition and nonpoint sources due to various human activities. The red tide features in the Bohai Sea seemed to be potentially related to the long-term nutrient variations.

1. Introduction

Nutrients are very important to primary productivity, which is the basis of marine ecosystems. The addition of nutrients to aquatic environments will stimulate the preferential growth of certain phytoplankton species under suitable conditions (Ferreira et al., 2007) and lead to further changes in their communities (Justić et al., 1995; Yunev et al., 2007). With the accelerated development of agriculture and industry over the past four decades, nutrient inputs from land have undergone significant changes globally, and these changes have resulted in variations in the concentrations and compositions of nutrients in seawater (Danielsson et al., 2008; Jickells, 1998; Paerl, 2006; Yu & Shen, 2011). Due to the potential effects of nutrient variations

©2018. American Geophysical Union. All Rights Reserved.

Journal of Geophysical Research: Oceans



Figure 1. Schematic illustration of the sites where historical data were collected in the Bohai Sea. The dots with "A1–A10" represent the sample sites from 1978 to 2016; the dotted lines represent the rivers.

on marine ecosystems, harmful algal blooms (Anderson et al., 2002; Glibert & Burkholder, 2011; Paerl, 2006; Yu & Shen, 2011; Yunev et al., 2007), jellyfish blooms (Purcell, 2012; Shin-Ichi, 2008; Xian et al., 2005), hypoxia, and ocean acidification may occur in coastal oceans (Cai et al., 2011; Kemp et al., 2009; Rabalais et al., 2002, 2014).

The Bohai Sea is a continental sea in China, which has a total area of 77,284 km² and an average depth of 18 m (Zhang et al., 2006). The Bohai Sea is composed of three bays (Liaodong Bay, Bohai Bay, and Laizhou Bay) and a central area, and it is connected to the northern Yellow Sea via the Bohai Strait (Figure 1). Approximately 40 rivers, including four major rivers (the Yellow River, Hai River, Daliao River, and Luan River), flow into the area and transport large amounts of nutrients. With the rapid expansion of industry, agriculture, and mariculture developments, nutrient fluxes from rivers (Zhang, 1996; Zhang et al., 1994), atmospheric deposition (Liu et al., 2008), wastewater discharge (Liu, 2006; Wang et al., 2016), and other sources have substantially increased. As an important spawning ground for fish and shrimps, the Bohai Sea has abundant marine resources and provides very important economic and ecological functions. Shandong Province, Liaoning Province, Hebei Province, and Tianjin municipality are located along the coast of the Bohai Sea, making it one of the most populated areas in China. The Bohai Sea is a relatively closed environment that is vulnerable to anthropogenic stress. Under the influence of dense human activities combined with changes in the natural environment, variations in the concentrations and compositions of nutrients in the Bohai Sea may lead to eutrophication and consequent ecological disasters. In recent decades, the frequency and affected area of red tides have increased significantly in the Bohai Sea (Shen, 2001; Zhao, 2010; Zou et al., 1985). Moreover, the variation in nutrients may boost phytoplankton production and algae blooms in the Bohai Sea, which may further enhance zooplankton production and supply more food to jellyfish, thus resulting in jellyfish blooms in the Bohai Sea (Dong et al., 2010; Purcell, 2012; Shin-Ichi, 2008; Xian et al., 2005). Hypoxia and coastal acidification (Yu & Zhang, 2016; Zhai et al., 2012; Zhang et al., 2016) have also been observed in the Bohai Sea since 2011. These ecological disasters in recent years have severely threatened the health of local marine ecosystems, and the variations in nutrients and the ecological environment of the Bohai Sea have raised increasing attention (Liu et al., 2008; Ning et al., 2010; Wu et al., 2013; Yu et al., 2000).

To examine the nutrient variations in the Bohai Sea, a study by Liu et al. (2008) showed that from 1959 to 1999, the annual mean concentration of dissolved inorganic nitrogen (DIN) and the N/P ratio increased, while the concentration of dissolved silicate (SiO₃-Si) and the Si/N ratio decreased; the concentration of dissolved inorganic phosphate (PO_4 -P) slightly increased from 1959 to 1982 and decreased from 1982 to 1999, which was also proven by the results of Cui and Song (1996). Similar increases in the DIN concentration



and N/P ratio and similar decreases in the PO₄-P and SiO₃-Si concentrations from the early 1980s to the late 1990s were also found in the studies by Yu et al. (2000) and Ning et al. (2010). However, the variation patterns of the nutrient concentrations in the Bohai Sea over the last 40 years are not the same as those observed in other previous studies, and ambiguity likely exists. In contrast to the continuous decreasing trends of the PO₄-P and SiO₃-Si concentrations from the early 1980s to the late 1990s in the aforementioned studies, Li et al. (2003) and Jiang et al. (2005) observed decreases from 1982 to 1992 and increases from 1992 to 1998 in the SiO₃-Si and PO_4 -P concentrations in the Bohai Sea. This result was supported by the results of the study by Zhang et al. (2006) from 1985 to 1998, which showed a decrease before 1990 and an increase after 1990 in the PO₄-P concentration. In addition, the results of the study by Li et al. (2003) showed that the DIN concentration in the Bohai Sea decreased from 1982 to 1992 and then increased from 1992 to 2000, whereas the results of the study by Xu et al. (2010) based on previous studies found that the DIN concentration increased from 1980 to 1995 and decreased from 1995 to 2005. These two distinct patterns of the variations in DIN in the Bohai Sea are not consistent with the aforementioned continuously increasing trend. Evidence has shown that discharge from the Yellow River (Cui et al., 1994; Li et al., 2003; Liu et al., 2012; Xu et al., 2010; Zhao et al., 2002) and atmospheric deposition (Cui, 2008; Liu et al., 2008; Shou et al., 2018; Zhang et al., 2004) are important sources of nutrients in the Bohai Sea, whereas on-land sewage treatment and chemical fertilizers may be other sources of DIN and PO₄-P (Li et al., 2003; Liu et al., 2005; Xu et al., 2010). As the ratios of different nutrients in the Bohai Sea changed with the variations in nutrient concentrations, it was believed that a gradual change from nitrogen limitation to phosphate and silicate limitations occurred during the early 1980s to the late 1990s (Jiang et al., 2005; Yu et al., 2000), and a large variance in the phytoplankton community structure was observed (Liu et al., 2008; Wu et al., 2013; Xu et al., 2010). According to the study by Yu et al. (2000), the increased N/P ratio and the decreased Si/N ratio from 1982 to 1999, which are potentially attributed to the increase in fertilizer and decrease in the flow of the Yellow River into the Bohai Sea, may be the major factors driving the occurrence of red tide dinoflagellates (Pyrrophyta) in the Bohai Sea. These previous studies primarily applied data from discrete years (i.e., discontinuous time series such as 1982, 1992, and 1998) to represent the long-term variations in nutrients in the Bohai Sea during different periods and focused on the variation trends before 2000. To date, the patterns of the long-term variations in the nutrients in the Bohai Sea remain controversial and unclear, especially the continuous variation trends after 2000. In this paper, a retrospective analysis of the long-term variation patterns of nutrients and their compositions in the Bohai Sea was performed based on historical data from 1978 to 2016. Seasonal differences were also considered by comparing the trends in summer and winter. This paper constitutes a considerable expansion of previous work, and we also explored potential influencing factors to reveal the typical regional response of the Bohai Sea to external forcing and its relationship to changes occurring in the ecosystem during this period.

2. Materials and Methods

2.1. Data Sources

Ten sampling sites A1-A10 are located on a transect across the Bohai Sea from southwest to northeast within the range of 119.13°E to 121.48°E longitude and 38.07°N to 40.45°N latitude (Figure 1). Site A1 between Bohai Bay and Laizhou Bay in the southwestern Bohai Sea is close to the Yellow River Estuary, whereas site A10 in Liaodong Bay in the northeastern Bohai Sea is close to the Daliao River Estuary. The transect survey data used in this study from 1978 to 2006 and from 2015 to 2016 were obtained from the State Oceanic Administration of China, and the data from 2013 to 2014 were obtained from the Yellow Sea Fisheries Research Institute, Chinese Academy of Fishery Sciences. The surveys were conducted every season (February, May, August, and November) and almost every year from 1978 to 1995 and 2013 to 2016, but only in winter (February/January) and summer (August/July) from 1996 to 2006. We selected the data from the winter and summer seasons to ensure that the data were comparable and the time series was continuous. The survey parameters included dissolved nitrate (NO₃-N), nitrite (NO₂-N), ammonia (NH₄-N), phosphate (PO₄-P), and silicate (SiO₃-Si). DIN was the sum of NO₃-N, NO₂-N, and NH₄-N. However, NO₃-N and NO₂-N measurements began in 1984, and NH₄-N measurements began in 1990; no nitrogen measurements were collected from 1996 to 1999, and SiO₃-Si was not measured from 2007 to 2016. The reference data for nutrients in the summer of 2008 (Liu et al., 2011) and the reference data for PO_4 -P in the summers of 2007-2011 (Shi, 2013) that were collected along the same transect were adopted to fill the data gaps during this period. All sampling and analytical methods followed the Chinese national seawater analysis standards, which ensured that the data were comparable throughout the study period.

In addition to the aforementioned survey data, the water discharge and sediment load data from the Yellow River at the Lijin station (obtained from the Yellow River Water Resources Commission and Yuan et al., 2006) were also analyzed to obtain the riverine inputs. The fertilization data for the Bohai Rim, which were obtained from China Statistical Yearbook (the National Bureau of Statistics of the People's Republic of China, 1978–2016), were analyzed to reveal the human impacts on the river inputs. The annual precipitation data from Tianjin (obtained from the National Bureau of Statistics of the People's Republic of China, 1978–2016 and Li et al., 2010) were used to reflect the wet deposition in the Bohai Sea. The annual frequencies of sand and dust events in northern China (obtained from the China Meteorological Administration, National Climate Committee, China Climate Bulletin) were used to reflect the dry deposition in the Bohai Sea. The data on wastewater discharge into the Bohai Sea were obtained from China Marine Statistical Yearbook (the State Oceanic Administration, 1985–2016). The data on mariculture area in the Bohai Sea were obtained from China Fishery Statistical Yearbook (Bureau of Fisheries of the Ministry of Agriculture, 1979-2016). Moreover, the red tide frequency and affected area in the Bohai Sea (obtained from the China Marine Statistical Yearbook and State Oceanic Administration, People's Republic of China, Chinese Marine Disaster Bulletins) were applied to analyze the potential ecological effects.

2.2. Data Analyses

The mean values of the nutrient concentrations at the surface layers and the bottom layers (i.e., 2 m above the seabed) of the 10 sites along the transect were calculated in both the winter and summer seasons, and the mean values of the two seasons were calculated as the annual means. Thus, the interannual variations in the surface and bottom nutrients for each component along the transect were obtained (1978–2016 for PO₄-P, 1978–2006 for SiO₃-Si, 1984–2014 for NO₃-N and NO₂-N, 1990–2014 for NH₄-N, and 1990–2016 for DIN), and linear regression analytical methods (including simple linear regression and binomial regression) were used to evaluate the variation trends. However, during the period from 2007 to 2012, no survey data were obtained from the transect in summer and winter. In the summer of 1993 and the winter of 1984, no parameters were measured; in the winter of 1990, NH₄-N was not measured, which indicated that the NH₄-N time series in winter began in 1991; in 2015 and 2016, the DIN and PO₄-P concentrations were measured only at the surface, and thus, no bottom data were analyzed. Due to the limitations of the different nutrient measurements stated above, the long-term analysis of the variations in N/P ratios was limited to the summers from 1990 to 2016 and winters from 1991 to 2016. For Si/N, the summers from 1990 to 2008 and winters from 1991 to 2006 were analyzed, and for Si/P, the summers from 1978 to 2008 and winters from 1978 to 2006 were analyzed.

3. Results

3.1. Variations in Nutrient Concentrations

In summer, the surface and bottom concentrations of PO_4 -P generally exhibited decreasing trends, whereas the concentrations of SiO₃-Si decreased from 1978 to 1987 and showed an increasing trend from 1987 to 2008 (Figures 2a–2f). The surface and bottom concentrations of DIN exhibited an overall increasing trend from 1990 to 2016 with a significant increase from 2002 to 2008 and subsequently maintained high values. The surface and bottom concentrations of NO₃-N showed synchronous increasing trends with the increasing DIN concentrations, whereas the concentrations of NH₄-N and NO₂-N showed moderate increasing trends. For the vertical differences, the bottom concentrations of nutrients in summer were slightly higher than the surface concentrations. The proportions of NO₃-N showed a significant increasing trend, while the proportions of NH₄-N and NO₂-N showed decreasing trends (Figures 2g–2h). In the summers from 1990 to 2000, NH₄-N accounted for the largest proportion of DIN, NO₃-N accounted for the second largest proportion, and NO₂-N contributed the least. However, as the proportion of NO₃-N was slightly larger at the surface than near the bottom; the opposite pattern was observed for the proportion of NH₄-N.

In winter, the surface and bottom concentrations of PO_4 -P fluctuated with a slightly increasing trend from 1978 to 2016 (Figures 3a–3f). Sharp decreases were found in 1987 and 1997. The SiO₃-Si concentrations in





Figure 2. Temporal variations in nutrient concentrations along the A1–A10 transect in the Bohai Sea in summers from 1978–2016: (a) surface PO_4 -P concentration, (b) bottom PO_4 -P concentration, (c) surface SiO_3 -Si concentration, (d) bottom SiO_3 -Si concentration, (e) surface dissolved inorganic nitrogen (DIN) concentration, (f) bottom DIN concentration, (g) surface proportions of different DIN forms, and (h) bottom proportions of different DIN forms. The dotted lines represent the variation trend lines determined by linear regression (including simple linear regression and binomial regression).





Figure 3. Temporal variations in nutrient concentrations along the A1–A10 transect in the Bohai Sea in winters from 1978–2016: (a) surface PO_4 -P concentration, (b) bottom PO_4 -P concentration, (c) surface SiO_3 -Si concentration, (d) bottom SiO_3 -Si concentration, (e) surface dissolved inorganic nitrogen (DIN) concentration, (f) bottom DIN concentration, (g) surface proportions of different DIN forms, and (h) bottom proportions of different DIN forms. The dotted lines represent the variation trend lines determined by linear regression (including simple linear regression and binomial regression).

winter showed a trend that was similar to that observed in summer, decreasing from 1978 to 1987 and increasing from 1987 to 2008. The surface and bottom concentrations of NO_3 -N exhibited uniform increasing trends from 1991 to 2016, and the concentrations of DIN showed synchronous increasing trends. The concentrations of NH_4 -N and NO_2 -N fluctuated over relatively steady ranges. The vertical differences in nutrient concentrations were less pronounced in winter than in summer. Moreover, the nutrient concentrations in winter tended to be higher than those in summer. The variation patterns of the proportions of different DIN forms were almost the same in the surface and bottom layers of the Bohai Sea in the winters from 1991 to 2014 (Figures 3g–3h). In the winters during this period, the proportions of NO_3 -N were the largest and showed a continuous increasing trend. The proportions of NH_4 -N were the second largest but much smaller than those of NO_3 -N. The proportions of NO_2 -N were the smallest and remained at approximately 3%.

The annual mean concentrations of PO_4 -P generally exhibited a slight decreasing trend, and the variations in the annual mean concentrations of SiO_3 -Si and DIN were similar to those in summer and winter (Figures 4a–4f). The bottom concentrations of PO_4 -P and SiO_3 -Si were slightly higher than the surface concentrations, while the concentrations of DIN exhibited the opposite patterns. In terms of the DIN composition at the surface and bottom of the Bohai Sea during the study period, the average proportions of NO_3 -N were the highest, those of NH_4 -N were the second highest, and those of NO_2 -N were the lowest (Figure 4g). The differences between the proportions of NO_3 -N and NH_4 -N were 2.43% at the surface and 6.40% at the bottom in summer and 66.49% at the surface and 65.21% at the bottom in winter. The proportions of NO_3 -N were higher in winter than in summer as well as throughout the year, while the NH_4 -N and NO_2 -N proportions were lower in winter. The proportions of NO_3 -N were greater at the surface than at the bottom in different seasons and throughout the year.

3.2. Variations in Nutrient Ratios

In summer, the surface and bottom ratios of N/P exhibited increasing trends, with rapid increases after 2002 (Figure 5a), and these changes are similar to the variation patterns of the DIN concentrations. The N/P ratios were also higher than the Redfield ratio of 16:1 after the 2000s. The Si/N ratio showed an overall decreasing trend and tended to be lower than the Redfield ratio of 1:1 from 2004 (Figure 5d). The Si/P ratio showed an overall increasing trend, especially after 1987 (Figure 5g), which is similar to the variation pattern of the SiO₃-Si concentration. The bottom ratios tended to be lower than the ratios at the surface. The patterns of the variations in the nutrient ratios in summer in the Bohai Sea indicated that the nutrient limitation condition changed from a nitrogen limitation to phosphorus and silicon limitations, which may have influenced the uptake and utilization of phytoplankton in the Bohai Sea. The overall trend of the variations in the nutrient ratios in winter generally followed the variation patterns observed in summer (Figures 5b, 5e, and 5h). The nutrient ratios in winter were much lower than those in summer, and the slopes of the variations in winter were much smaller than those in summer. The variations in the annual mean nutrient ratios generally followed the patterns in summer (Figures 5c, 5f, and 5i). The annual mean nutrient ratios were lower at the bottom than at the surface, while the vertical differences in the N/P and Si/N ratios were small.

4. Discussion

4.1. Comparison With Previous Studies

During the period from the early 1980s to the late 1990s, an increase in the DIN concentration and a decrease in the PO_4 -P concentration were observed in the Bohai Sea in three studies using 1982-1992-1998 time series (Cui & Song, 1996; Liu et al., 2008; Yu et al., 2000) and one study using a continuous time series from 1978 to 1996 (Ning et al., 2010). Our study revealed the same variation patterns of the annual mean concentrations of DIN and PO_4 -P during the same period. The decreasing trend of the PO_4 -P concentration was also observed by Shi (2013), who analyzed the PO_4 -P variation in the Bohai Sea in the summers from 1978 to 2012. Moreover, according to the continuous increase in the DIN concentration from 1985 to 1998 in the study by Zhang et al. (2006) and the similar increase starting from 1959 in the studies by Cui and Song (1996) and Liu et al. (2008), the increasing trend of the DIN concentration can be traced back to 1959; based on the results of this study, particularly the DIN variation after 2000, we suggest an increasing trend of the





100





Figure 5. Temporal variations in nutrient ratios along the A1–A10 transect in the Bohai Sea in summers, winters, and throughout the years from 1978–2016: (a) N/P ratio in summer, (b) N/P ratio in winter, (c) annual mean N/P ratio, (d) Si/N ratio in summer, (e) Si/N ratio in winter; (f) annual mean Si/N ratio, (g) Si/P ratio in summer, (h) Si/P ratio in winter, and (i) annual mean Si/P ratio. The dotted lines represent the variation trend lines determined by linear regression (including simple linear regression and binomial regression).

DIN concentration in the Bohai Sea from 1959 to 2016, with a rapid increase after 2002. Similarly, according to the results of by Cui and Song (1996) and Liu et al. (2008), the PO_4 -P concentration slightly increased and maintained a high level from 1959 to the early 1980s; based on our results of the PO_4 -P variation from 1978 to 2016, we suggest an overall decreasing pattern of the PO_4 -P concentration in the Bohai Sea from 1959 to 2016, with a slight increase during the period from 1959 to 1982.

According to the results of the SiO₃-Si variation in the Bohai Sea from Cui and Song (1996), Yu et al. (2000), and Liu et al. (2008) using discontinuous (1959-)1982-1992(-1998) time series and the results from Ning et al. (2010) analyzing the continuous variation from 1978 to 1996, the SiO₃-Si concentration tended to decrease from 1982 to 1998. However, the overall decreasing pattern of SiO₃-Si from 1978 to 1996 observed by Ning et al. (2010) included a decreasing period from 1978 to 1987 and an increasing period from 1978 to 1996, which is roughly consistent with the variation patterns observed by Li et al. (2003) and Jiang et al. (2005) using a 1982-1992-1998 time series and by Zhang et al. (2006) using a continuous time series from 1985 to 1998. In accordance with the decrease in SiO₃-Si from 1959 to 1982 (Liu et al., 2008), the decreasing trend of the SiO₃-Si concentration can be traced back to 1959; based on our results, we suggest that the SiO₃-Si concentration in the Bohai Sea decreased from 1959 to 1987 and increased from 1987 to 2008.

It is likely that the adoption of a discontinuous time series (e.g., the 1982-1992-1998 series in Yu et al., 2000; Li et al., 2003; and Jiang et al., 2005) and the trend over different seasons (e.g., May and October in Yu et al., 2000, and May, August, and October in Jiang et al., 2005) may have resulted in the differences and inaccuracies in some previous studies that analyzed the long-term variations in nutrients in the Bohai Sea. The seasonal differences are also proven in this study. We find that the nutrient concentrations in the Bohai Sea are much lower in summer than in winter (Figures 2 and 3): the PO_4 -P concentrations in summer range from [0.040, 0.595] (µmol/L), whereas the concentrations in winter range from [0.114, 1.119] (µmol/L); the SiO_3 -Si concentrations in summer range from [0.207, 27.387] (µmol/L), whereas the concentrations in winter range from [0.658, 37.482] (µmol/L); and the DIN concentrations in summer range from [1.183, 14.217] (µmol/L), whereas the concentrations in winter range from [5.275, 21.320] (µmol/L). The variation patterns of nutrients also show seasonal differences (Figures 2 and 3): the PO₄-P concentration in summer decreases from 1978 to 2016, whereas the concentration in winter shows an overall increasing trend; the DIN and NO₃-N concentrations in summer show significant rapid increases after 2002, whereas the increasing patterns in winter tend to be uniform; the increasing trends of the NH₄-N and NO₂-N concentrations in summer are also different from the relatively steady concentration levels in winter. Moreover, the proportions of different DIN forms differ in the different seasons (Figures 2 and 3). The proportion of NH_4 -N tends to be higher in summer than in winter, especially from 1990 to 2000, when NH₄-N accounted for the largest proportion of DIN. The proportion of NO₃-N was much higher in winter than in summer and contributed an average of more than 80% of the total DIN during the study period. Therefore, it is strongly recommended that the nutrient conditions should be measured in more than one season in a year, and the data that are measured in different seasons should be averaged to eliminate the seasonal differences to obtain the long-term variation patterns of the nutrients in the Bohai Sea.

Consistent with the results of Yu et al. (2000), Jiang et al. (2005), and Ning et al. (2010), the increasing N/P ratio and decreasing Si/N ratio in our results indicate an evolutionary trend from an earlier nitrogen limitation to the potential phosphorus and silicon limitations from 1978 to 2016 in the Bohai Sea. The Si/P variation from our study roughly corresponds to the interdecadal results of Jiang et al. (2005), who observed a decrease from 1982 to 1992 and a substantial increase from 1992 to 1998. In addition, the increasing trend of the Si/P ratio after 1990, which is similar to the variation in the SiO₃-Si concentration, and the overall decreasing trend of the Si/N ratio are also consistent with the trends observed in the southern Yellow Sea (Wei et al., 2015).

4.2. Main Nutrient Sources

In this section, the long-term variations in riverine inputs, atmospheric deposition, and nonpoint sources under natural changes and anthropogenic impacts are discussed as the main factors (Li, 2010; Mackas & Harrison, 1997) that influence the long-term variations in the nutrients in the Bohai Sea.

4.2.1. The Variations in Riverine Inputs

In coastal and estuarine areas, riverine inputs are likely to account for a majority of the total nutrient sources. Studies have shown that changes in riverine nutrient loads may shift the distribution and composition of nutrients in coastal continental shelf areas worldwide, thus exerting effects on coastal ecosystems (Humborg et al., 1997; Liu et al., 2005; Rabalais et al., 1996; Zhang, 1996).

As the main water body that receives flow from more than 40 rivers, the environmental conditions of the Bohai Sea may be greatly impacted by the discharge from the surrounding rivers. Rivers may play an important role in nutrient transport from land to the semiclosed Bohai Sea. The Yellow River is the second longest river in China and contributes 64.3% of the freshwater discharge from all rivers flowing into the Bohai Sea (Liu, 2006). Thus, the Yellow River was analyzed as the main riverine source of nutrients to the Bohai Sea in this study. In recent decades, there has been a decreasing trend of water and sediment fluxes from rivers into the Bohai Sea due to the impacts of natural changes and human activities, such as the construction of dams and reservoirs and diversion irrigation from mainstreams (Wang, Yang, et al., 2006, 2007). The water discharge and sediment load of the Yellow River from 1978 to 2002 also showed an overall decreasing trend that was probably due to the construction of hydropower stations during this period because the shrinkage in water discharge was particularly sharp after the river closure at Longyangxia at the end of 1979, the reservoir impoundment of the Longyangxia hydropower station in October 1986, and the river closure at the Xiaolangdi hydropower station in October 1997 (Figure 6a). According to the statistics, the drying up days





Figure 6. (a). Long-term variations in water discharge and sediment load of the Yellow River at the Lijin station. The data were obtained from the Yellow River Water Resources Commission and Yuan et al. (2006). (b) Long-term variations in fertilization in the Bohai Rim. The data were obtained from the National Bureau of Statistics of the People's Republic of China. (c–e). Plots of the annual mean nutrient concentrations versus salinity at the surface at site A1 near the Yellow River Estuary: (c) SiO₃-Si, (d) PO₄-P, and (e) DIN. (f–h). Plots of the annual mean nutrient concentrations at site A1 versus the annual sediment load of the Yellow River at the Lijin station: (f) SiO₃-Si, (g) PO₄-P, and (h) DIN.

of the Yellow River at the Lijin station reached a maximum of 226 days in 1997 compared to 8 days in 1980 (Yellow River Water Resources Commission). The water discharge and sediment load increased significantly in 2002 when the water sediment regulations came into operation (Figure 6a), and drying up has not occurred in the Yellow River since 2002 (Yellow River Water Resources Commission). From 2003 to 2013, the water discharge of the Yellow River showed an increasing trend, while the sediment load continued to decrease (Figure 6a).

conditions) on Wiley Online Library for rules of use; OA articles

are governed by the applicable Creativ

commons Licens

2169291, 2019, 1, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2018JC014765 by Utrecht University Library, Wiley Online Library on [17/08/2023]. See the Terms and Conditions (https://onlinelibrary.wiley



Table	1
1 4010	-

Relative Contributions of the Main Nutrient Sources to the Bohai Sea in 2011

	Riverine input			Atmospheric	Nonpoint sources	
%	Yellow River (excluding particulate nutrients)	Yellow River (particulate nutrients)	Total riverine inputs (excluding particulate nutrients)	deposition (dry + wet)	Wastewater discharge	Mariculture
DIN PO ₄ -P SiO ₃ -Si	3.93 0.12 55.17	20.76	4.76 0.48 95.39	25.81 0.22 4.61	61.23 71.34	8.20 7.20

Note. The relative contributions were approximated as the proportions of the nutrient fluxes from a nutrient source to the nutrient fluxes from all nutrient sources listed. The riverine nutrient fluxes were estimated using the water discharge data of the Yellow River at the Lijin station in 2011, which were obtained from the Yellow River Water Resources Commission, and the nutrient concentrations of the Yellow River at the Lijin station, which were obtained from Chen et al. (2013). The total particulate phosphorus flux was calculated based on the sediment load at the Lijin station from the Yellow River Water Resources Commission and the particulate phosphorus content from He et al. (2010). The total riverine nutrient fluxes (excluding particulate nutrients) were derived from Liu et al. (2011). The DIN flux from atmospheric deposition was calculated from Shou (2017), while the PO_4 -P and SiO_3 -Si fluxes were calculated from Zhang et al. (2004). The nitrogen and phosphorus fluxes from the wastewater discharge into the Bohai Sea were estimated from the National Bureau of Statistics of the People's Republic of China and the unit nutrient emissions from Liu (2006).

As the fresh water of the Yellow River flows into the Bohai Sea, it also carries large amounts of sediments and nutrients that significantly change the nutrient concentrations in the estuarine and adjacent sea areas (Liu, 2015; Liu et al., 2012). The Yellow River transports approximately 410×10^9 mol/year of dissolved ions, while its chemical erosion rate is close to the world average (Zhang et al., 1995). Weathering and erosion accelerated by anthropogenic activities, such as farming in recent decades, may be the main reason for the increases in many chemical element fluxes, including silicates (Hu et al., 1982; Tréguer et al., 1995; Zhang et al., 1995, 2015). The cumulative effects of continued global warming and acidification also favor weathering and erosion in drainage basins (Wei et al., 2015). Previous studies have shown that the significant decreases in the water flux and sediment content of the Yellow River before 2000 resulted in a decreasing trend of the SiO₃-Si concentration in the Bohai Sea (Ning et al., 2010; Yu et al., 2000), and the SiO₃-Si from stream discharge may account for up to 80% of the total SiO₃-Si inputs (Liu et al., 2005). In this study, the variation trend of the SiO₃-Si concentration in the Bohai Sea roughly corresponded to the variation in the riverine water discharge from the Yellow River to the Bohai Sea from 1978 to 2008 (Figures 4c, 4d and 6a). In particular, the sharp decreases in the SiO₃-Si concentration in 1987 and 1997 corresponded to the river closure and reservoir impoundment of hydropower stations. Based on the correlation analyses between the nutrient concentrations and salinity near the Yellow River Estuary during our study period (Figures 6c–6e), the SiO₃-Si concentration was observed to have a fine negative correlation coefficient, which indicated that the fresh water from the Yellow River may have an important impact on the SiO₃-Si concentration. Moreover, the relative contribution of SiO₃-Si from the Yellow River was estimated to be more than 50% of the total sources into the Bohai Sea, and the total riverine input was estimated to be more than 95% (Table 1). Therefore, riverine input is suggested to be an important factor for the long-term SiO₃-Si variations in the Bohai Sea.

Compared to other rivers worldwide, the ratio of sediment load to water discharge is very high in the Yellow River (Zhang et al., 1995). Evidence has shown that phosphorus is transported to the ocean primarily in particulate phases via riverine flux (Conley et al., 1995; Guo et al., 2004; Meng et al., 2015). Dissolved PO_4 -P adsorbs to particles such as amorphous ferric hydroxides in freshwater and is then released as salinity increases during the process of freshwater–seawater mixing (Carritt & Goodgal, 1954; Lebo, 1990). Evidence has shown that the DIN and PO_4 -P fluxes into the Bohai Sea through rivers tended to increase from 1979 to 1990 and decrease from 1990 to the early 2000s (Liu, 2006), which does not entirely correspond to the nutrient variation patterns in the Bohai Sea. In addition, we found the annual mean concentration of PO_4 -P in the Bohai Sea roughly matched the overall decreasing trend in the sediment loads of the Yellow River, and the sharp decreases in 1987 and 1997 corresponded to the decreases in the sediment loads of the Yellow River owing to the river closures (Figures 4a and 4b). Moreover, it was observed that the PO_4 -P concentration near the Yellow River Estuary and the sediment load were positively correlated (Figure 6g), and the contribution of PO_4 -P from the Yellow River increased from 0.12% to 20.88% of the total sources into the Bohai Sea if the particulate contribution was considered (Table 1). Thus, the difference in the estimated riverine PO_4 -P flux and the PO₄-P concentration in the Bohai Sea may arise from the substantial release of particulate phosphorus from sediment, which is underestimated; the riverine inputs, especially the internal mechanisms involving sediment transport, could be at least somewhat important to the long-term variations in PO_4 -P in the Bohai Sea. However, the PO₄-P fluxes from the Yellow River into the Bohai Sea that were estimated by Liao et al. (2013) showed an increasing trend from 2002 to 2012, which is contrary to the decreasing trend of the sediment load of the Yellow River during the same period. This phenomenon was also observed in the southern Yellow Sea (Wei et al., 2015), and was attributed to the increasingly extensive fertilizer use. The DIN variation in the Bohai Sea did not match the variations in the water discharge and sediment load of the Yellow River (Figures 4e, 4f and 6a), and the DIN concentration near the Yellow River Estuary was also observed to be uncorrelated with salinity (Figure 6e). According to Cui (2008), improper fertilization led to annual losses of 2.41×10^4 t of nitrogen and 0.61×10^4 t of phosphorus in the three provinces and one municipality in the Bohai Rim. Moreover, nutrient losses due to natural soil losses in the amounts of 0.1×10^4 t of nitrogen and 0.08×10^4 t of phosphorus should be added. A significant increasing trend in the compound fertilizer (mainly including nitrogen, phosphorus, and potassium) used in the Bohai Rim was found, while the amounts of nitrogen fertilizer and phosphate fertilizer remained steadily high from 1987 to 2016 (Figure 6b). The large inputs of fertilizer into the drainage basins as well as the increasing trend of total fertilization in recent decades might partially compensate for the deficits of DIN and PO₄-P caused by the decreased river discharge and sediment load as well as the trap effect (i.e., the absorption of nutrients by phytoplankton in the reservoir and sediment precipitation; Wei et al., 2015) and might even promote increases in the DIN and PO₄-P fluxes from rivers.

4.2.2. The Variations in Atmospheric Deposition

In addition to riverine inputs, atmospheric deposition is also proven to be an important pathway of nutrient inputs to seas and may provide from 10% to more than 40% of the nitrogen in N-limited estuarine and coastal waters (Okin et al., 2011; Paerl et al., 2002; Zhang et al., 1999). Globally, nitrogen from atmospheric deposition to the ocean reached 3.5×10^8 t/year (Wollast, 1991). Evidence has shown that atmospheric deposition is likely an important source of DIN to the Bohai Sea (Liu et al., 2008; Zhang et al., 2004). In 1998, 26180 t of nitrogen was input into the Bohai Sea via atmospheric deposition (Cui, 2008). The relative contribution of DIN from atmospheric deposition was estimated to be larger than that from riverine input (Shou et al., 2018). The contribution of DIN from atmospheric deposition was estimated to be 25.81% of the total DIN input into the Bohai Sea in 2011 (Table 1), which indicated that the variation in atmospheric deposition may be somewhat important to the long-term DIN variations in the Bohai Sea.

Windblown dusts play an important role in marine ecosystems once the dusts are deposited and dissolved (Tu et al., 2015). Approximately 63.9% of sand and dust events likely affect seas in China (Zhang et al., 2005), and the deposition fluxes of particulate phosphorus during the dust storm events associated with precipitation were approximately 500-1,000 times greater than the daily average flux during nondust days in the Yellow Sea (Shi et al., 2013). Dusts moving from the Loess Plateau and Mongolian Plateau were observed to be deposited into the Bohai Sea (Zhang et al., 2005). In 1998, 770 t of phosphorus was input into the Bohai Sea via atmospheric deposition (Cui, 2008). Tianjin is one of the major cities on the periphery of the Bohai Sea. Therefore, the long-term variations in the trends of dust and precipitation in this region can partially reflect the effects of dust events and wet deposition on the Bohai Sea. A study on the dust in this region indicated that the number of days with dust weather in Tianjin decreased from the mid-1960s to 1999, with an abrupt increase in the early 2000s (Duan et al., 2005). The annual frequencies of sand and dust weather in northern China showed a decreasing trend from 2000 to 2017 (Figure 7a), which was consistent with the decreasing trend of the PO₄-P concentration in the Bohai Sea during the same period (Figure 4). However, the contribution of PO₄-P from atmospheric deposition was very limited compared with the contributions from rivers and nonpoint sources (Table 1), which indicated that atmospheric deposition may not be the main factor determining the long-term variation pattern of the PO₄-P in the Bohai Sea. Observational data showed that the overall precipitation in the Tianjin region exhibited a decreasing trend before the late 1990s and a gradually increasing trend from the late 1990s to 2015 (Figure 7a), which roughly corresponded to the long-term variations in the SiO_3 -Si concentration in the Bohai Sea (Figure 4). Although the SiO_3 -Si flux directly from atmospheric deposition to the Bohai Sea was estimated to be only 4.61% of the total input flux (Table 1), the long-term variations in the annual precipitation affect the Earth surface processes around the sea that are related to wet deposition, and large-scale precipitation may promote water and soil losses

Journal of Geophysical Research: Oceans



Figure 7. (a) Long-term variations in the annual sand and dust weather frequencies in northern China. The data were obtained from the China Meteorological Administration, National Climate Committee, China Climate Bulletin. (b) Long-term variations in the annual precipitation in Tianjin. The data were obtained from the National Bureau of Statistics of the People's Republic of China, China Statistical Yearbook and Li et al. (2010).

(Wei et al., 2015). Terrestrial SiO_3 -Si induced by the scouring and erosion effects of rainfall on the land surface may enter the ocean, thereby increasing the nutrient concentrations in the ocean (Wei et al., 2015). **4.2.3. The Variations in Nonpoint Sources**

In addition to the above sources, the variations in nonpoint nutrient sources should also be considered, such as the increasing discharge of industrial, agricultural, and domestic wastewater and the expanding area of mariculture. With the intensification of human activities, a large amount of nutrients have been input into the sea, and the anthropogenic inputs of nutrients will alter the concentrations and composition ratios of nutrients in the sea. These changes will greatly impact the absorption and utilization of nutrients by different phytoplankton species in the local ecosystem.

According to the investigation of nonpoint sources of pollution from land to sea in 2017, 9,600 nonpoint pollution sources (drain outlets) were found along the coast of China (http://www.soa.gov.cn/xw/ztbd/ztbd 2017/2017wthzxdc/mtbd/201801/t20180118_60021.html), which indicates that there was a source every 2 km along the coastline on average. The number of pollution sources along the coastline of the Bohai Sea was estimated to be 640. From 1985 to 2015, the total wastewater discharge continuously increased, the domestic sewage discharge showed the same increasing trend, and the industrial wastewater discharge was maintained at approximately 4×10^9 t with no significant variations (Figure 8a). The study by Xu et al. (2010) indicated the significant influences of sewage disposal on the nutrients in the Bohai Sea. Evidence has shown that wastewater transport leads to increased concentrations of ammonium and phosphate in the surface waters of the Bohai Sea, and the phosphate from wastewater accounts for 31% of the total inputs, whereas the ammonium from wastewater accounts for 84% of the total inputs (Liu et al., 2005). The variations in the trend of the total wastewater discharge induced by increased domestic sewage corresponded to the long-term trend of the DIN concentration in the Bohai Sea (Figures 8a and 4), and the contribution of nitrogen from wastewater was estimated to be 61.23% of the total sources into the Bohai Sea (Table 1). Therefore, it is indicated that the dense population and high volumes of domestic sewage discharge are likely important factors accounting for the rapidly increased DIN concentration in the Bohai Sea.

Expansion of the aquaculture industry has resulted in increasing nutrients in its effluents (Boyd et al., 2007). Fish or other cultured species retain only approximately 20–30% of the nitrogen from their feed and a smaller fraction of phosphorus, while the remaining nutrients are released into the water as dissolved and solid waste (Krom & Neori, 1989; Lupatsch & Kissil, 1998). The contributions of nitrogen and phosphorus from mariculture were estimated to be 8.20% and 7.20% of the total sources into the Bohai Sea, respectively (Table 1). The mariculture area in the Bohai Sea increased from 1979 to 2016 (Figure 8b), which also corresponded to the long-term variations in the DIN concentration in the Bohai Sea. Evidence has shown that aquaculture has a greater influence on nitrogen nutrients than on phosphate nutrients (Kang & Xu, 2016). Therefore, the continuously expanded area used for mariculture in the Bohai Sea could contribute to the increase in the inputs of DIN nutrients.



Journal of Geophysical Research: Oceans



Figure 8. (a). Long-term variations in wastewater discharge into the Bohai Sea. The data were obtained from the State Oceanic Administration, China Marine Statistical Yearbook. (b). Long-term variations in the mariculture area in the Bohai Sea. The data were obtained from the Bureau of Fisheries of the Ministry of Agriculture, China Fishery Statistical Yearbook.

4.3. Potential Effects on Ecological Disasters

The reasons for the occurrence of ecological disasters in the Bohai Sea may be complex. An increasing trend of chlorophyll *a* content from 2002 to 2006 was believed to be caused by increased nutrient concentrations (Xu et al., 2010). As mentioned in section 3.1, the surface nutrient concentrations were significantly lower than the bottom concentrations in summer (Figure 2). During this period, nutrients at the surface of the Bohai Sea were substantially utilized by phytoplankton that were in vigorous growth stages (Cui et al., 1994; Zhou et al., 2017). However, in winter, this phenomenon was mitigated, and there were much smaller differences between the surface and bottom concentrations of nutrients (Figure 3).

Similarly, the increase in red tides is likely a response to the change in nutrients. The frequency and area of red tide events in the Bohai Sea tended to increase after 1989 and reached an explosive phase after 1998 when the affected area significantly expanded and the frequency greatly increased (Figure 9a). From 1990 to 1995, the low number of red tide outbreaks in the Bohai Sea matched the low DIN concentrations and N/P ratios during the same period (Figure 4). In addition, during the explosive phase of red tides in the Bohai Sea, the dominant species also changed from single Pyrrophyta species to a more diverse composition, including Pyrrophyta, Bacillariophyta, Rhaphidophyta, Chrysophyta, Cryptophyta, and Cyanophyta (Figure 9a), and the percentage of toxic algal blooms also increased (State Oceanic Administration, People's Republic of China, Chinese Marine Disaster Bulletins). During this phase, the DIN concentration increased significantly, and the SiO₃-Si concentration increased slightly, while the PO₄-P concentration decreased (Figure 4). Notably, the variations in the Si/N ratio during the period from 1990 to 2008 roughly matched the variations in the frequency, area, and species diversity of red tides during this period (Figures 5f and 9a): the decreasing Si/N ratio in the early 1990s corresponded to lower red tide frequencies, and the decreasing Si/N ratio in the 2000s also corresponded to lower red tide frequencies and lower dominant species diversities accompanied by lower ranks of Bacillariophyta as well as larger red tide areas. In addition, in the years when red tide occurred, the DIN concentrations and the N/P and Si/N ratios tended to be higher than those in the years without red tide occurrences; for the SiO₃-Si and PO₄-P concentrations, the differences were not prominent (Figures 9b and 9c). Analysis of the potential relationships between the red tides and nutrients in the Bohai Sea (Figures 9d-9m) revealed that the red tide area and frequency tended to increase when the DIN and PO₄-P concentrations increased. When the SiO₃-Si concentration was lower than approximately 10 μ mol/L, the red tide area and frequency tended to increase as SiO₃-Si increased; when the concentration was greater than 10 µmol/L, decreasing trends of the red tide area and frequency were observed. When the N/P ratio was lower than approximately 30, the red tide area and frequency tended to increase as the N/P





Figure 9. (a). Long-term variations in the red tide frequency and area in the Bohai Sea. The data were obtained from the China Marine Statistical Yearbook and State Oceanic Administration, People's Republic of China, Chinese Marine Disaster Bulletins. The Si/N ratios in the figure are 4 times the real values to provide a better comparison. The dominant species listed in the figure are listed in the order of their affected areas in a year. (b and c). Nutrient concentrations and ratios in years (b) without red tide occurrence and (c) with red tide occurrence in the Bohai Sea. (d-m). The relationships between red tides and nutrients in the Bohai Sea: (d) red tide area versus DIN concentration, (e) red tide frequency versus DIN concentration, (f) red tide area versus N/P ratio, (g) red tide frequency versus SiO₃-Si concentration, (i) red tide frequency versus SiO₃-Si concentration, (i) red tide area versus Si/N ratio, (l) red tide area versus PO₄-P concentration, and (m) red tide frequency versus PO₄-P concentration. The dotted blue lines in the figures represent fitting curves.

ratio increased; when the ratio was greater than 30, the red tide area and frequency decreased as the N/P ratio increased. The red tide frequency showed increasing trends as the Si/N ratio increased in the ranges of (0, 2.5) and (2.5, 5), and the area showed a potential decreasing trend as the Si/N ratio increased. Generally, under nonlimiting conditions, diatoms are fast-growing species and have a higher growth rate than flagellates (Banse, 1982). High nutrient levels and sufficient silicon supplies will promote the preferential growth of diatoms (Wang, Qi, et al., 2006), and Pyrrophyta such as Noctiluca scintillans will then feed on the enhanced diatoms; during these blooms, substantial amounts of nutrients are consumed (Feng, 2004; Yin et al., 2013). As the nitrogen concentration was generally in excess due to increasing anthropogenic impacts in recent decades, the N/P ratio tended to increase and phosphorus limitation was likely to occur, especially after the consumption of phosphorus by phytoplankton, and during this phase, the proportion of diatoms may decrease by up to 50% (Fu et al., 2012). The increasing N/P ratio in the Bohai Sea during our study period, which was higher than the Redfield ratio after the 2000s, and the decreasing Si/N ratio, which was lower than the Redfield ratio after 2003 (Figure 5), as well as the variations in nutrient concentrations (Figure 4), indicated an evolutionary trend from an early nitrogen limitation to the current potential phosphorus and silicon limitations in the Bohai Sea. This trend may limit the growth of diatoms and promote the development of Pyrrophyta in the Bohai Sea if other conditions are suitable (Yu et al., 2000). It is therefore suggested that the changes in nutrient concentrations and nutrient limitations may be a potential factor accounting for the increasing occurrence of red tides (in terms of area, frequency, and diversity of dominant species) in the Bohai Sea.

5. Conclusions

Nutrients are very important to marine ecosystems as they support the growth of phytoplankton. Due to the accelerated development of agriculture and industry over the past four decades, nutrient inputs from land have experienced significant changes in China, which has resulted in variations in the concentrations and compositions of nutrients in marginal seas. As the only enclosed sea in China, the Bohai Sea is a relatively closed environment that is particularly vulnerable to human activities. In this paper, the long-term variation patterns of nutrients and their compositions in the Bohai Sea, as well as their potential influencing factors, were analyzed based on historical summer and winter data from 1978 to 2016.

The results showed that in summer, the DIN concentration in the Bohai Sea continuously increased from 1990 and showed a more rapid increase after 2002. The PO₄-P concentration exhibited a decreasing trend from 1978 to 2016. The SiO_3 -Si concentration decreased from 1978 to 1987 and increased from 1987 to 2008. In winter, the PO_4 -P concentration fluctuated with a slight increasing trend, while the concentrations of SiO₃-Si and DIN exhibited increasing trends. The variations in the NO₃-N concentrations were consistent with the variations in the DIN concentrations in summer and winter, while the NH₄-N and NO₂-N concentrations increased in summer but fluctuated over relatively steady ranges in winter. For the vertical difference, the nutrient concentrations were higher at the bottom than at the surface in summer, whereas the difference was not significant in winter. In summer and winter, the N/P ratio increased with a rapid increase after 2002, the Si/N ratio decreased, and the Si/P ratio showed an overall slightly increasing trend. The bottom ratios tended to be lower than the surface ratios in summer, whereas the vertical differences were not prominent in winter. The nutrient ratios were much lower in winter than in summer. The nutrient limitation changed from nitrogen limitation to phosphorus and silicon limitations, which might influence the uptake and utilization of nutrients by phytoplankton in the Bohai Sea. Based on the results of previous studies from 1959 and the results of the present study from 1978 to 2016, there was an overall increase in the annual mean concentration of DIN from 1959 to 2016, an overall decrease in the annual mean concentration of PO₄-P from 1978 to 2016, a decrease in the annual mean concentration of SiO₃-Si from 1959 to 1987, and an increase in the SiO₃-Si concentrations from 1987 to 2008. The dominant factors determining the long-term variations in the DIN concentrations in the Bohai Sea could be atmospheric deposition and nonpoint sources due to various human activities; the primary factors determining the variations in the PO₄-P concentration could be the variations in riverine inputs and nonpoint sources related to natural changes and human activities; riverine inputs have the greatest influence on the variations in the SiO₃-Si concentration. The long-term nutrient variations in the Bohai Sea have potential ecological impacts on the frequencies, affected areas, and diversity of the dominant species in the local red tide blooms.



Acknowledgments

This study was funded by Aoshan Talents Program supported by the Scientific and Technological Innovation Project of the Qingdao National Laboratory for Marine Science and Technology (2016ASKJ02), the National Key Research and Development Program of China (2017YFC1404402), the Fundamental Research Funds for the Central Universities (201822006). We thank our laboratory colleagues for their collaboration in sampling and data acquisition. The detailed survey data used in this study are presented as supporting information.

References

- Anderson, D. M., Glibert, P. M., & Burkholder, J. M. (2002). Harmful algal blooms and eutrophication: Nutrient sources, composition, and consequences. *Estuaries*, 25(4), 704–726. https://doi.org/10.1007/BF02804901
- Banse, K. (1982). Cell volumes, maximal growth rates of unicellular algae and ciliates, and the role of ciliates in the marine pelagial. *Limnology and Oceanography*, 27(6), 1059–1071. https://doi.org/10.4319/lo.1982.27.6.1059
- Boyd, C. E., Tucker, C., Mcnevin, A., Bostick, K., & Clay, J. (2007). Indicators of resource use efficiency and environmental performance in fish and crustacean aquaculture. *Reviews in Fisheries Science*, 15(4), 327–360. https://doi.org/10.1080/10641260701624177
- Bureau of Fisheries of the Ministry of Agriculture (1979-2016). China Fishery Statistical Yearbook.
 Cai, W., Hu, X., Huang, W., Murrell, M. C., Lehrter, J. C., Lohrenz, S. E., et al. (2011). Acidification of subsurface coastal waters enhanced by eutrophication. *Nature Geoscience*, 4(11), 766–770. https://doi.org/10.1038/ngeo1297
- Carritt, D. E., & Goodgal, S. (1954). Sorption reactions and some ecological implications. Deep Sea Research, 1(4), 224–243. https://doi.org/ 10.1016/0146-6313(54)90005-2
- Chen, P., Liu, S., Zhang, G., Li, L., Gu, P., & Cao, X. (2013). Monthly variation of nutrient concentrations and fluxes in the lower Huanghe River: Under the influence of artificial floods (in Chinese with English abstract). *Acta Oceanologica Sinica*, *35*(2), 59–71. https://doi.org/ 10.3969/j.issn.0253-4193.2013.02.008
- Conley, D. J., Smith, W. M., Cornwell, J. C., & Fisher, T. R. (1995). Transformation of particle-bound phosphorus at the land-sea interface. *Estuarine, Coastal and Shelf Science*, 40(2), 161–176. https://doi.org/10.1016/S0272-7714(05)80003-4
- Cui, Y., & Song, Y. (1996). Study on evaluation of nutrient status in the Bohai Sea (in Chinese with English abstract). Marine Fisheries Research, 17(1), 57-62.

Cui, Y., Yang, Q., & Song, Y. (1994). Distribution of phosphate and dissolved oxygen and their relation in the Bohai Sea in summer (in Chinese with English abstract). Marine Environmental Science, 13(4), 31–35.

- Cui, Z. (2008). Study on scheme of total emission control of main chemical pollutants in 13 cities around Bohai Sea (in Chinese with English abstract), (Doctoral dissertation). Ocean University of China.
- Danielsson, A., Papush, L., & Rahm, L. (2008). Alterations in nutrient limitations—Scenarios of a changing Baltic Sea. Journal of Marine Systems, 73(3–4), 263–283. https://doi.org/10.1016/j.jmarsys.2007.10.015
- Dong, Z., Liu, D., & Keesing, J. K. (2010). Jellyfish blooms in China: Dominant species, causes and consequences. Marine Pollution Bulletin, 60(7), 954–963. https://doi.org/10.1016/j.marpolbul.2010.04.022
- Duan, L., Liu, A., & Guo, J. (2005). Variation regularities and wavelet transform features of dust storms in spring in Tianjin (in Chinese with English abstract). *Meteorological Science and Technology*, 33(6), 513–515. https://doi.org/10.19517/j.1671-6345.2005.06.008

Feng, J. (2004). The study of HABs ecosystem dynamics and prediction in Bohai Bay (in Chinese with English abstract), (Doctoral dissertation), Tianjin University.

- Ferreira, J. G., Bricker, S. B., & Simas, T. C. (2007). Application and sensitivity testing of a eutrophication assessment method on coastal systems in the United States and European Union. *Journal of Environmental Management*, 82(4), 433–445. https://doi.org/10.1016/j. jenvman.2006.01.003
- Fu, M., Wang, Z., Pu, X., Xu, Z., & Zhu, M. (2012). Changes of nutrient concentrations and N: P: Si ratios and their possible impacts on the Huanghai Sea ecosystem. Acta Oceanologica Sinica, 31(4), 101–112. https://doi.org/10.1007/s13131-012-0224-x
- Glibert, P. M., & Burkholder, J. A. M. (2011). Harmful algal blooms and eutrophication: "Strategies" for nutrient uptake and growth outside the Redfield comfort zone. *Chinese Journal of Oceanology and Limnology*, 29(4), 724–738. https://doi.org/10.1007/s00343-011-0502-z
- Guo, L., Zhang, J., & Guéguen, C. (2004). Speciation and fluxes of nutrients (N, P, Si) from the upper Yukon River. *Global Biogeochemical Cycles*, *18*, GB1038. https://doi.org/10.1029/2003GB002152
- He, H., Yu, Z., Yao, Q., Chen, H., & Mi, T. (2010). The hydrological regime and particulate size control phosphorus form in the suspended solid fraction in the dammed Huanghe (Yellow River). *Hydrobiologia*, 638(1), 203–211. https://doi.org/10.1007/s10750-009-0041-1
- Hu, M., Stallard, R. F., & Edmond, J. M. (1982). Major ion chemistry of some large Chinese rivers. *Nature*, 298(5874), 550–553. https://doi.org/10.1038/298550a0
- Humborg, C., Ittekkot, V., Cociasu, A., & Bodungen, B. V. (1997). Effect of Danube River dam on Black Sea biogeochemistry and ecosystem structure. Nature, 386(6623), 385–388. https://doi.org/10.1038/386385a0
- Jiang, H., Cui, Y., Chen, B., Chen, J., & Song, Y. (2005). The variation trend of nutrient salts in the Bohai Sea (in Chinese with English abstract). Marine Fisheries Research, 26(6), 61–67. https://doi.org/10.3969/j.issn.1000-7075.2005.06.012
- Jickells, T. D. (1998). Nutrient biogeochemistry of the coastal zone. Science, 281(5374), 217–222. https://doi.org/10.1126/ science.281.5374.217
- Justić, D., Rabalais, N. N., & Turner, R. E. (1995). Stoichiometric nutrient balance and origin of coastal eutrophication. Marine Pollution Bulletin, 30(1), 41–46. https://doi.org/10.1016/0025-326X(94)00105-I
- Kang, P., & Xu, S. (2016). The impact of mariculture on nutrient dynamics and identification of the nitrate sources in coastal waters. Environmental Science & Pollution Research, 23(2), 1300–1311. https://doi.org/10.1007/s11356-015-5363-0
- Kemp, W. M., Testa, J. M., Conley, D. J., Gilbert, D., & Hagy, J. D. (2009). Temporal responses of coastal hypoxia to nutrient loading and physical controls. *Biogeosciences*, 6(12), 2985–3008. https://doi.org/10.5194/bg-6-2985-2009
- Krom, M. D., & Neori, A. (1989). A total nutrient budget for an experimental intensive fishpond with circularly moving seawater. *Aquaculture*, *83*(3-4), 345–358. https://doi.org/10.1016/0044-8486(89)90045-8

Lebo, M. E. (1990). Phosphate uptake along a coastal plain estuary. Limnology and Oceanography, 35(6), 1279–1289. https://doi.org/ 10.4319/lo.1990.35.6.1279

- Li, C., Liu, D., & Huang, H. (2010). Characteristics of precipitation and precipitation days from 1958 to 2007 in Tianjin (in Chinese with English abstract). *Journal of Meteorology and environment*, 26(4), 8–11. https://doi.org/10.3969/j.issn.1673-503X.2010.04.002
- Li, L. (2010). Effects of exchange fluxes of nutrients at the sediment and water interface and the Huanghe input on nutrient dynamics of the Bohai Sea (in Chinese with English abstract), (Master's dissertation). Ocean University of China.
- Li, Z., Bai, J., Shi, J., & Gao, H. (2003). Distributions of inorganic nutrients in the Bohai Sea of China. Journal of Ocean University of Qingdao (Oceanic and Coastal Sea Research), 2(1), 112–116. https://doi.org/10.1007/s11802-003-0038-8
- Liao, W., Zhang, L., Chen, H., Xiao, C., & Zhang, X. (2013). Nutrient variation and fluxed estimation in the Yellow River Estuary from 2001 to 2011 (in Chinese with English abstract). *Periodical of Ocean University of China*, 43(1), 81–86. https://doi.org/10.16441/j.cnki. hdxb.2013.01.012
- Liu, J. (2006). Study on the Main Pollutant Fluxes Flowing into the Bohai (in Chinese with English abstract), (Master's dissertation). Ocean University of China.

- Liu, S. (2015). Response of nutrient transports to water-sediment regulation events in the Huanghe basin and its impact on the biogeochemistry of the Bohai. Journal of Marine Systems, 141, 59–70. https://doi.org/10.1016/j.jmarsys.2014.08.008
- Liu, S., Li, L., Zhang, G., Liu, Z., Yu, Z., & Ren, J. (2012). Impacts of human activities on nutrient transports in the Huanghe (Yellow River) estuary. Journal of Hydrology, 430-431(1), 103–110. https://doi.org/10.1016/j.jhydrol.2012.02.005

Liu, S., Li, L., & Zhang, Z. (2011). Inventory of nutrients in the Bohai. Continental Shelf Research, 31(16), 1790–1797. https://doi.org/ 10.1016/j.csr.2011.08.004

- Liu, S., Zhang, J., Chen, H., & Zhang, G. (2005). Factors influencing nutrient dynamics in the eutrophic Jiaozhou Bay, North China. Progress in Oceanography, 66(1), 66–85. https://doi.org/10.1016/j.pocean.2005.03.009
- Liu, S., Zhang, J., Gao, H., & Liu, Z. (2008). Historic changes in flux of matter and nutrient budgets in the Bohai Sea. Acta Oceanologica Sinica, 27(5), 81–97. https://doi.org/10.3969/j.issn.0253-505X.2008.05.006
- Lupatsch, I., & Kissil, G. W. (1998). Predicting aquaculture waste from gilthead seabream (Sparus aurata) culture using a nutritional approach. Aquatic Living Resources, 11(4), 265–268. https://doi.org/10.1016/S0990-7440(98)80010-7
- Mackas, D. L., & Harrison, P. J. (1997). Nitrogenous nutrient sources and sinks in the Juan de Fuca Strait/Strait of Georgia/Puget Sound Estuarine System: Assessing the potential for eutrophication. *Estuarine, Coastal and Shelf Science, 44*(1), 1–21. https://doi.org/10.1006/ecss.1996.0110
- Meng, J., Yu, Z., Yao, Q., Bianchi, T. S., Paytan, A., Zhao, B., et al. (2015). Distribution, mixing behavior, and transformation of dissolved inorganic phosphorus and suspended particulate phosphorus along a salinity gradient in the Changjiang Estuary. *Marine Chemistry*, 168, 124–134. https://doi.org/10.1016/j.marchem.2014.09.016

National Bureau of Statistics of the People's Republic of China (1978-2016). China Statistical Yearbook.

- Ning, X., Lin, C., Su, J., Liu, C., Hap, Q., Le, F., & Tang, Q. (2010). Long-term environmental changes and the responses of the ecosystems in the Bohai Sea during 1960-1996. Deep Sea Research Part II Topical Studies in Oceanography, 57(11-12), 1079–1091. https://doi.org/ 10.1016/j.dsr2.2010.02.010
- Okin, G. S., Baker, A. R., Tegen, I., Mahowald, N. M., Dentener, F. J., Duce, R. A., et al. (2011). Impacts of atmospheric nutrient deposition on marine productivity: Roles of nitrogen, phosphorus and iron. *Global Biogeochemical Cycles*, 25, GB2022. https://doi.org/10.1029/ 2010GB003858
- Paerl, H. W. (2006). Assessing and managing nutrient-enhanced eutrophication in estuarine and coastal waters: Interactive effects of human and climatic perturbations. *Ecological Engineering*, *26*(1), 40–54. https://doi.org/10.1016/j.ecoleng.2005.09.006
- Paerl, H. W., Dennis, R. L., & Whitall, D. R. (2002). Atmospheric deposition of nitrogen: Implications for nutrient over-enrichment of coastal waters. *Estuaries*, 25(4), 677–693. https://doi.org/10.1007/BF02804899
- Purcell, J. E. (2012). Jellyfish and ctenophore blooms coincide with human proliferations and environmental perturbations. Annual Review of Marine Science, 4(1), 209–235. https://doi.org/10.1146/annurev-marine-120709-142751
- Rabalais, N. N., Cai, W. J., Carstensen, J., Conley, D. J., Fry, B., Hu, X., et al. (2014). Eutrophication-driven deoxygenation in the coastal ocean. Oceanography, 27(1), 172–183. https://doi.org/10.5670/oceanog.2014.21
- Rabalais, N. N., Turner, R. E., Justić, D., Dortch, Q., Wiseman, W. J., & Gupta, B. K. S. (1996). Nutrient changes in the Mississippi River and system responses on the adjacent continental shelf. *Estuaries*, 19(2), 386–407. https://doi.org/10.2307/1352458
- Rabalais, N. N., Turner, R. E., & Scavia, D. (2002). Beyond science into policy: Gulf of Mexico hypoxia and the Mississippi River: Nutrient policy development for the Mississippi River watershed reflects the accumulated scientific evidence that the increase in nitrogen loading is the primary factor in the worsening of hypoxia in the northern Gulf of Mexico. *Bioscience*, 52(2), 129–142. https://doi.org/10.1641/ 0006-3568(2002)052[0129:BSIPGO]2.0.CO;2
- Shen, Z. (2001). Historical changes in nutrient structure and its influences on phytoplantkon composition in Jiaozhou Bay. Estuarine, Coastal and Shelf Science, 52(2), 211–224. https://doi.org/10.1006/ecss.2000.0736
- Shi, J., Zhang, J., Gao, H., Tan, S., Yao, X., & Ren, J. (2013). Concentration, solubility and deposition flux of atmospheric particulate nutrients over the Yellow Sea. Deep-Sea Research Part II, 97, 43–50. https://doi.org/10.1016/j.dsr2.2013.05.004
- Shi, Q. (2013). Spatial-temporal changes of the annual phosphate in the Bohai Sea in summer. *Marine Science Bulletin*, 32(4), 395–402. https://doi.org/10.11840/j.issn.1001-6392.2013.04.006
- Shin-ichi, U. (2008). Blooms of the giant jellyfish *Nemopilema nomurai*: A threat to the fisheries sustainability of the East Asian Marginal Seas. *Plankton & Benthos Research*, *3*(Supplement), 125–131. https://doi.org/10.3800/pbr.3.125
- Shou, W. (2017). Contribution of atmospheric deposition to the nutrients and its ecological effects in the Bohai (in Chinese with English abstract), (Doctoral dissertation), East China Normal University.
- Shou, W., Zong, H., Ding, P., & Hou, L. (2018). A modelling approach to assess the effects of atmospheric nitrogen deposition on the marine ecosystem in the Bohai Sea, China. *Estuarine, Coastal and Shelf Science, 208*, 36–48. https://doi.org/10.1016/j.ecss.2018.04.025 State Oceanic Administration (1985–2016). China Marine Statistical Yearbook.
- Tréguer, P., Nelson, D. M., Van bennekom, A. J., DeMaster, D. J., Leynaert, A., & Quéguiner, B. (1995). The silica balance in the world ocean: A re-estimate. *Science*, 268(5209), 375–379. https://doi.org/10.1126/science.268.5209.375
- Tu, Q., Hao, Z., & Pan, D. (2015). Mass deposition fluxes of Asian dust to the Bohai Sea and Yellow Sea from geostationary satellite MTSAT: A case study. Atmosphere, 6(11), 1771–1784. https://doi.org/10.3390/atmos6111771
- Wang, H., Yang, Z., Saito, Y., Liu, J., & Sun, X. (2006). Interannual and seasonal variation of the Huanghe (Yellow River) water discharge over the past 50 years: Connections to impacts from ENSO events and dams. *Global and Planetary Change*, 50(3–4), 212–225. https://doi. org/10.1016/j.gloplacha.2006.01.005
- Wang, H., Yang, Z., Saito, Y., Liu, J., Sun, X., & Wang, Y. (2007). Stepwise decreases of the Huanghe (Yellow River) sediment load (1950– 2005): Impacts of climate change and human activities. *Global and Planetary Change*, 57(3-4), 331–354. https://doi.org/10.1016/j. gloplacha.2007.01.003
- Wang, N., Li, H., Lu, Y., & Zhang, J. (2016). Study on the relations between economic development of Bohai-Zone and the quality of sewage into the sea (in Chinese with English abstract). *Transactions of Oceanology and Limnology*, 5, 28–36. https://doi.org/10.13984/j.cnki. cn37-1141.2016.05.005
- Wang, Z., Qi, Y., Chen, J., Xu, N., & Yang, Y. (2006). Phytoplankton abundance, community structure and nutrients in cultural areas of Daya Bay, South China Sea. *Journal of Marine Systems*, 62(1-2), 85–94. https://doi.org/10.1016/j.jmarsys.2006. 04.008
- Wei, Q., Yao, Q., Wang, B., Wang, H., & Yu, Z. (2015). Long-term variation of nutrients in the southern Yellow Sea. Continental Shelf Research, 111, 184–196. https://doi.org/10.1016/j.csr.2015.08.003
- Wollast, R. (1991). The coastal organic carbon cycle: Fluxes, sources and sinks. In R. F. C. Mantoura, J. M. Martin, & R. Wollast (Eds.), Ocean margin processes in global change, Dahlem Workshop Reports (pp. 365–381). Chichester: Wiley Interscience.



- Wu, Z., Yu, Z., Song, X., Yuan, Y., Cao, X., & Liang, Y. (2013). Application of an integrated methodology for eutrophication assessment: A case study in the Bohai Sea. *Chinese Journal of Oceanology and Limnology*, *31*(5), 1064–1078. https://doi.org/10.1007/s00343-013-2286-9
 Xian, W., Kang, B., & Liu, R. (2005). Jellyfish blooms in the Yangtze Estuary. *Science*, *307*(5706), 41c. https://doi.org/10.1126/science.307.5706.41c
- Xu, S., Song, J., Li, X., Yuan, H., Li, N., Duan, L., & Sun, P. (2010). Changes in nitrogen and phosphorus and their effects on phytoplankton in the Bohai Sea. *Chinese Journal of Oceanology and Limnology*, 28(4), 945–952. https://doi.org/10.1007/s00343-010-0005-3
- Yin, C., Zhang, Q., Zou, T., Zhang, C., Cui, J., & Sun, J. (2013). Analysis for *Noctiluca Scintillans* red tide in Bohai Bay. *Transacions of Oceanology and Liminology*, *2*, 99–104. (In Chinese with English abstract). https://doi.org/10.13984/j.cnki.cn37-1141.2013.02.008
- Yu, J., & Zhang, H. (2016). Spatial distribution and isotopic signature of organic carbon and nitrogen in surface sediments from Bohai Sea. Paper (Abstract) Presented at Ocean Sciences Meeting, American Geophysical Union.
- Yu, Z., Mi, T., Xie, B., Yao, Q., & Zhang, J. (2000). Changes of the environmental parameters and their relationship in recent twenty years in the Bohai Sea (in Chinese with English abstract). *Marine Environmental Science*, 19(1), 15–19. https://doi.org/10.3969/j.issn.1007-6336.2000.01.004
- Yu, Z., & Shen, Z. (Eds.) (2011). Eutrophication in the Changjiang River Estuary and Adjacent waters (in Chinese with English abstract) (p. 549). Beijing: Science Press.
- Yuan, Z., Chu, Z., & Yang, Y. (2006). Estuary runoff and amount of transported sediments of Yellow River and ecological environments of Yellow River Delta (in Chinese with English abstract). *Journal of Palaeogeography*, 8(1), 125–130. https://doi.org/10.7605/ gdlxb.2006.01.013
- Yunev, O. A., Carstensen, J., Moncheva, S., Khaliulin, A., Ærtebjerg, G., & Nixon, S. (2007). Nutrient and phytoplankton trends on the western Black Sea shelf in response to cultural eutrophication and climate changes. *Estuarine, Coastal and Shelf Science*, 74(1-2), 63–76. https://doi.org/10.1016/j.ecss.2007.03.030
- Zhai, W., Zhao, H., Zheng, N., & Xu, Y. (2012). Coastal acidification in summer bottom oxygen-depleted waters in northwestern-northern Bohai Sea from June to August in 2011. *Chinese Science Bulletin*, *57*(9), 1062–1068. https://doi.org/10.1007/s11434-011-4949-2
- Zhang, H., Li, Y., Tang, C., Zou, T., Yu, J., & Guo, K. (2016). Spatial characteristics and formation mechanisms of bottom hypoxia zone in the Bohai Sea during summer (in Chinese with English abstract). *Chinese Science Bulletin*, 61(14), 1612–1620. https://doi.org/10.1360/ N972015-00915
- Zhang, J. (1996). Nutrient elements in large Chinese estuaries. Continental Shelf Research, 16(8), 1023–1045. https://doi.org/10.1016/0278-4343(95)00055-0
- Zhang, J., Huang, W., Létolle, R., & Jusserand, C. (1995). Major element chemistry of the Huanghe (Yellow River), China—Weathering processes and chemical fluxes. Journal of Hydrology, 168(1–4), 173–203. https://doi.org/10.1016/0022-1694(94)02635-O

Zhang, J., Huang, W., Liu, M., & Cui, J. (1994). Eco-social impact and chemical regimes of large Chinese rivers—A short discussion. Water Research, 28(3), 609–617. https://doi.org/10.1016/0043-1354(94)90011-6

Zhang, J., Yu, Z., Raabe, T., Liu, S., Starke, A., Zou, L., et al. (2004). Dynamics of inorganic nutrient species in the Bohai seawaters. *Journal of Marine Systems*, 44(3-4), 189–212. https://doi.org/10.1016/j.jmarsys.2003.09.010

Zhang, J., Zhang, Z., Liu, S., Wu, Y., Xiong, H., & Chen, H. (1999). Human impacts on the large world rivers: Would the Changjiang (Yangtze River) be an illustration? *Global Biogeochemical Cycles*, *13*, 1099–1105. https://doi.org/10.1029/1999GB900044

- Zhang, K., Gao, H., Zhang, R., Zhu, Y., & Wang, Y. (2005). Sources and movement routes of sans-dust aerosols and their impact probabilities on China seas in 2000–2002 (in Chinese with English abstract). Advances in Earth Science, 20(6), 627–636. https://doi.org/ 10.3321/j.issn:1001-8166.2005.06.005
- Zhang, Q., Jin, Z., Zhang, F., & Xiao, J. (2015). Seasonal variation in river water chemistry of the middle reaches of the Yellow River and its controlling factors. Journal of Geochemical Exploration, 156(3), 101–113. https://doi.org/10.1016/j.gexplo.2015.05.008
- Zhang, Z., Zhu, M., Wang, Z., & Wang, J. (2006). Monitoring and managing pollution load in Bohai Sea, P R China. Ocean and Coastal Management, 49(9-10), 706-716. https://doi.org/10.1016/j.ocecoaman.2006.06.005

Zhao, D. Z. (2010). The rules for occurrence of red tide in typical sea areas of China. Beijing: Ocean Press.

- Zhao, L., Wei, H., & Feng, S. (2002). Annual cycle and budgets of nutrients in the Bohai Sea (in Chinese with English abstract). Environmental Sciences, 23(1), 78–81. https://doi.org/10.13227/j.hjkx.2002.01.019
- Zhou, Y., Zhang, C., Shi, X., & Su, R. (2017). Distribution characteristics of chlorophyll a and its influencing environmental factors in Bohai Sea and Yellow Sea (in Chinese with English abstract). *China Environmental Science*, *37*(11), 4259–4265. https://doi.org/10.3969/j. issn.1000-6923.2017.11.031
- Zou, J., Dong, L., & Qin, B. (1985). Preliminary studies on eutrophication and red tide problems in Bohai Bay. *Hydrobiologia*, 127(1), 27–30. https://doi.org/10.1007/BF00004660